Author’s response to reviews

Title: A seven-year retrospective analysis of patch test data in a cohort of patients with contact dermatitis in Sri Lanka

Authors:
suramya keragala (suramya1983@yahoo.com)
tharuka herath (tharukaherath11@gmail.com)
TS keragala (TSkeragala@gmail.com)
M.A.M.H. Malavi (Malavi@gmail.com)
chaturaka rodrigo (c.rodrigo@unsw.edu.au)
chalukya gunesekara (chaluka@gmail.com)

Version: 5 Date: 15 Jun 2019

Author’s response to reviews:

1. Data Permission

Please state whether the data obtained for your study was publically availible under 'availability of data and materials'. If data is not publically available please state any permissions obtained to access the data.

done

2. Authors Contributions

-- Currently the initials used are confusing. Please ensure these match the authors listed on the title page.

done
3. Availability of Data and Materials
-- Please indicate who data can be obtained from, for example the corresponding author on reasonable request.
done

4. Consent for Publication
-- Please state “Not Applicable” in this section.
done

5. Additional Files
-- Please only include additional files that you wish to publish.

-- Please also add a section “Additional files” where you list the following information about your supplementary material:
- File name
- Title of data
- Description of data
added

Please ensure that all additional files are explicitly referred to in the main text.
done
6. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

done